

Delft University of Technology

A Multidisciplinary Review Into the Evolution of Risk Concepts and Their Assessment Methods

Xu, Yuanyuan; Reniers, Genserik; Yang, Ming

DOI 10.3390/pr12112449

Publication date 2024 **Document Version** Final published version

Published in Processes

Citation (APA)

Xu, Y., Reniers, G., & Yang, M. (2024). A Multidisciplinary Review Into the Evolution of Risk Concepts and Their Assessment Methods. *Processes*, *12*(11), Article 2449. https://doi.org/10.3390/pr12112449

Important note

To cite this publication, please use the final published version (if applicable). Please check the document version above.

Copyright Other than for strictly personal use, it is not permitted to download, forward or distribute the text or part of it, without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license such as Creative Commons.

Takedown policy

Please contact us and provide details if you believe this document breaches copyrights. We will remove access to the work immediately and investigate your claim.





Yuanyuan Xu^{1,*}, Genserik Reniers^{1,2,3} and Ming Yang¹

- ¹ Safety and Security Science Section, Faculty of Technology, Policy and Management, Delft University of Technology, 2628 BX Delft, The Netherlands
- ² Faculty of Applied Economics, Antwerp Research Group on Safety and Security (ARGoSS), Universiteit Antwerpen, B-2000 Antwerp, Belgium
- ³ Centre for Economics and Corporate Sustainability (CEDON), KULeuven, B-1000 Brussels, Belgium
- * Correspondence: y.xu-8@tudelft.nl

Abstract: Risk is a complex and multi-faceted concept defined and addressed differently across disciplines. Recent years have seen a growing recognition of the need for a more comprehensive and integrated risk perspective combining insights from various disciplines. However, those works have rarely been analyzed in literature reviews. This article explores the possible way of advancing how we assess and manage risks by reviewing the theories, approaches, models, and other fundamental aspects of different disciplines relevant to risk concepts. Additionally, we compare the origins, connections, and differences between state-of-the-art risk science and risk concept research in various disciplines. Some suggestions and future directions are provided for improving risk assessment. This paper helps to deepen the understanding of the risk concept and advance the development of the risk management arena.

Keywords: risk concept; multidisciplinary risk perspective; risk; risk fundamentals; risk assessment



Citation: Xu, Y.; Reniers, G.; Yang, M. A Multidisciplinary Review into the Evolution of Risk Concepts and Their Assessment Methods. *Processes* 2024, 12, 2449. https://doi.org/10.3390/ pr12112449

Academic Editor: Carlos Sierra Fernández

Received: 12 October 2024 Revised: 25 October 2024 Accepted: 4 November 2024 Published: 5 November 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

1. Introduction

The risk concept is the foundation of the approaches and methods of risk management, as terminology can affect and become a part of the theoretical framework in which we study risk and how we cope with it [1]. The practices of modern formal risk management have only emerged in recent decades, nevertheless, people have been dealing with risk problems for a long time. Since the 1990s, there has been increasing interest in redefining and refining the concept of risk, and substantial knowledge has been added to this fundamental area [2]. The distinction between risk and related concepts also received much attention (i.e., risk and hazard; risk and risk perception [3,4]. Aven's work provides a historical perspective, showing how the definition of risk has shifted in recent decades from probability-based thinking to an emphasis on uncertainty [5]. Boholm explored how Swedish government agencies define risk and identified inconsistencies between usage and definitions [6].

Modern society is increasingly occupied with preventing, managing, and mitigating all types of risk across a range of societal sectors, such as insurance, common law, natural disasters, epidemic disease, pollution, food contamination, building and fire codes, transportation accidents, and traffic injuries [7,8]. The literature on formal risk studies has expanded across multiple disciplines, each contributing unique insights on the taxonomy of risk perspectives. For instance, Jasanoff identified "two cultures" divide in risk analysis: disciplines such as mathematics, biostatistics, toxicology, and engineering, contribute to quantitative risk analysis on one side, while law, psychology, sociology, and economics contribute to nonquantitative side [9]. Althaus's multidisciplinary survey of risk literature uncovered conflicting attributes of risk across fields, with risk being viewed as objective or subjective, knowable or unknowable, individual or collective, and both rational and emotional [10]. Similarly, Renn's review highlights a distinct divide between the technical and natural sciences, which focus on the physical impacts of risk, and the economic, psychological, and social sciences, which emphasize the relationship between risk and society [11,12].

In recent years, formal risk management is increasingly recognized in organizations as a systematic approach to avoid losses and disasters. Many integrated risk management frameworks are developed to identify, assess and treat risks, from organizational context to national safety and security. These frameworks—such as the COSO Enterprise risk management framework [13]; the risk framework of the Dutch Government [14]; and the ISO 31000 standard on risk management [15], Cabinet Office [16]—cover a broad spectrum, including infectious diseases, ecological safety, cyber threats, major accidents, social and political stability, terrorism, international and military threats, economic, etc. Despite ongoing scientific debate on the multidimensional nature of risk, nearly all integrated frameworks are based on the rational idea that risk is a combination of probability/uncertainty and consequences, with little attention to contextual factors [17].

Nonetheless, many real-world risk problems are complex and multi-faceted, requiring a new risk management paradigm that bridges the disciplinary divide and allows for a more holistic approach to address the complexity of modern risks. While prior research has reviewed disciplinary perspectives on risk, there remains a need for studies that systematically analyze how definitions of risk are linked to disciplinary backgrounds, and how these different perspectives might collaborate effectively. This paper re-examines the concept of risk from a multidisciplinary perspective, with a focus on risk definitions and assessment methods. It is hoped that this review will:

- Mitigate the tendency to view and assess risk without considering its context;
- Clarify the reasons why risk is understood and assessed differently across disciplines;
- Shed light on the intellectual origins of current approaches to risk;
- Provide a foundation for future interdisciplinary collaboration.

The literature review is organized as follows. Section 2 elaborates on the literature review methodology of this study. Section 3 explores the historical development of risk research in different disciplines, with a focus on the initial engagement of different fields in the study of risk and main contributions. Section 4 identifies and categorizes diverse definitions and interpretations of risk. Section 5 explores "how" risk is assessed according to these varied definitions and interpretations. Before concluding, Section 6 discusses the impact of multidisciplinary perspectives on current risk theories and anticipates potential future directions in risk management.

2. The Review Methodology

The objective of this study is not to provide a thorough review to cover all risk concepts and its assessment methods. It is unrealistic to cover all relevant studies for such a broad topic. The primary aim is to establish the perspectives and methods of major disciplines in addressing risks. Snowballing is a literature review method that identify additional papers from reference list or citations [18]. We utilized a structured approach that integrates a backward snowballing literature review (from the reference lists) approach, complementary database search, and personal knowledge in this study. The detailed flowchart of snowballing literature review process is shown in Figure 1.

The included articles are limited to the period of publication from 1900 to September 2023. Articles not written in English are excluded. To avoid bias that some publishers may be in favor of certain fields of study as well as to avoid overlooking some obscure papers, Google Scholar is set as the main database. The literature review study is divided into two phases. It starts with the snowballing sampling method. Firstly, the initial highly relevant and highly cited papers are identified using keyword search ("risk definition" OR "risk concept" OR "risk assessment") from the Google Scholar database. Then, the identified initial known papers are included in the snowballing sampling process. The backward snowballing process is an iterative process involves several steps. First, based on the screening results, the highly cited papers were selected for further review by looking at

the title and abstract. Then, the references of the initial selected sample were reviewed and examined by reading the title and abstract. At last, the new set of relevant literature was included in another round of the snowballing process. The new set of literature underwent the same snowballing process until new papers did not contribute insight. To improve the after that, complementary database research is applied to identify articles that may have been omitted in the first phase. Finally, the identified articles are categorized by discipline for further characterization analysis.



Figure 1. Flowchart of the snowballing literature review process.

3. Historical Development of Risk Studies

Before diving into how risk concept is defined and assessed, it is essential to understand the origins and evolution of risk studies. A historical timeline that begins in 1600 is presented in Figure 2. The classification of disciplines mainly refers to Ref. [10]. Nine disciplines—mathematics, philosophy, economics, anthropology, psychology, sociology, law, engineering, and medicine—are discussed. The upper markers along the timeline indicate the sequence of initial engagement of different fields in the study of risk. The notes below the timeline mark significant risk theories, approaches, or phenomena in certain periods.





It should be noted that the aim of this section is to grasp a broad overview of how different disciplines approach risk, serving as a foundation for readers to understand the antecedents of various risk definitions and assessment methods. The in-depth discussions on perspectives of risk can be found in more specific studies on the topic.

3.1. Mathematics

Many documents [10,19,20] suggest that the concept of risk may have emerged from significant developments in social, intellectual, and economic life in the Middle Ages. In 1657, mathematician Blaise Pascal introduced probability theory. After that, the mathematical calculation of chance was soon applied to gambling and economic ventures, investments, and insurance [21]. This heuristic was further enhanced by Pascal's work in 1662, which proposed the maximizing expected value principle (agents facing risk would maximize the expected value) for calculating the fair price of a gamble [22]. Daniel Bernoulli would later build upon this insight in 1738, when he introduced the concept of utility and formulated the maximizing utility principle that explained why people would pay only a small price for a game of infinite mathematical expectation [23]. The probability theorems of 17th-century mathematicians provided the groundwork for the prevailing conceptions and practices of risk today [24].

3.2. Economics

The earliest discussion of risk in scientific economics evolved around the 1890s when research began to be undertaken on formulating a theory of distribution related to profit [25]. This sparked an interest in risk and risk-bearing, which manifested itself in the insurance literature. At the same time, applied mathematics sought to apply statistical methods to economics, resulting in the use of relative frequency as a measurement approach for risk. By 1920, the dispersion of a relative frequency distribution had been identified as risk [26]. In 1921, to make a distinction between measurable uncertainty and unmeasurable one, economist Frank Knight defined risk as "objective uncertainty" that is measurable when enough evidence is collected [27]. In the mid-1940s, building on Bernoulli's expected utility principle, John von Neumann and Oskar Morgenstern developed the Expected Utility (EU) theory [28]. Utility can be understood as benefit, advantage, pleasure, good, or happiness [29]. The Expected Utility (EU) theory is one of the most influential normative decision theories to specify how individuals make decisions under risk [30]. It is built upon utility theory—people value risks subjectively according to the utility value of gains and losses, which is related to ones' wealth status [31]. For instance, people who are risk averse will overestimate the loss while less sensitive to gain. Since the Second World War, the EU

theory has played a significant role in management (especially decision science), economics, and psychology [23].

3.3. Engineering

With the development of industrial society, a growing number of large and complex systems combined with operational risks gave rise to scientific risk research of engineering to systems safety [32]. Around 1924, the development of quantitative approaches to system safety and reliability in the Bell Telephone System constituted an important building block for the emergence of Probabilistic Risk Assessment (PRA) in various fields of complex engineering systems [33]. After World War II (1939–1945), the profound development of technologies, especially in the chemical and nuclear industries, brought about potential catastrophic and long-lasting consequences, resulting in systematic attempts to study societal risks [34]. This led to the establishment of modern risk management studies in 1955 [35]. Since the 1970s, as late modern societies began to pay more attention to the uncertainties accompanying the increasing ignorance and rapid development of modern technology and industry, risk research and risk management have become increasingly scientific issues [36]. The engineering sector has been influenced by the statistical method in finance of calculating probabilities to develop the idea of risk [37]. Engineering problems gave rise to a quantitative approach to system safety, the foundation of the emergence of quantitative risk assessment (QRA). QRA establishes connections between threats and consequences [38]. As a core method in engineering risk research, QRA was first applied to nuclear reactor risks in 1975 and has since been used in various non-nuclear situations [39]. Scientific methods for identifying causal links between adverse health effects and different types of hazardous activities and mathematical theories of probability jointly built the roots of modem risk analysis [7].

3.4. Medicine

The disease panorama in Western countries has changed considerably over the last century, moving from acute to more chronic disease, which caused a shift to more preventive medical activity [40]. Meanwhile, calculating risk entrenched in the scientific world provides the theoretical bases for health-related decision-making in clinical settings. Widespread techniques and methods for the analysis of "risk" in the professional literature on health and health care have developed since the 1960s. Health risk appraisal/assessment/management began to emerge in the medical field in that period [41]. Since the 1980s, "risk" has appeared more often in medical journals, especially in epidemiological literature. This phenomenon was defined by Skolbekken as a "risk epidemic" [42]. In evidence-based medicine, efforts are made to gather the data of risk factors and how they lead to the outcome. The emergence of health risk assessment has also brought about many controversies, such as how to avoid over-medicalization; human bodies are non-linear systems which cannot be measured by linear models that are risk-assessment-based.

3.5. Law

With the increasing moral anxiety of modern lawyers and the societal disappointment in their ethical regulation, legal systems underwent a dramatic change in the 1960s, and risk management norms gradually took over contemporary law and society [43]. Contemporary legislation developments turn to the preventive, future-oriented logic of risk. Modern law adopted the conception derived from economics that an action can be said to generate loss if it increases the probability of a loss [44]. The probabilistic process helps the legal system to adopt a broader and more expansive view of the sources of loss. It is devoted to risk control, with less focus on strict causation than on contribution to the occurrence of harm. The same trends could be seen in criminal law. Risk-driven law diminishes the appreciation of the moral choices facing lawyers in practice and the other-regarding obligations of lawyers in society [43]. Since the early 1980s, criminal law has attempted to act as a preventive force to control the risk of future crime. Risk-driven criminal law focuses on managing the risk of crime rather than the crime itself [45]. Sentencing offenders as dangerous, rather than sentencing them for a specific offense, appears to have increased in recent years [46].

3.6. Anthropology

Since the 1970s, more participation of anthropologists has shifted toward risks and hazards, as they are aware that populations' lives are affected by the environment [47]. Technological progress has brought about many more man-made risks to modern society than traditional societies, so the scientific community has become more aware that social concerns must be considered while promoting new technology. This initiative allows cultural and social anthropologists to delve into previously unexplored territory [48]. Culture is anthropology's leading innovation and core explanatory tool; it is proven to be the key to understanding why and how people worldwide deal with risks [21]. The concept of "culture" came to refer to the distinctions between thoughts and manners of behavior of people around the globe. In its deep and holistic framework, it incorporates kinship, subsistence, economics, politics, dress, food, exploitation of place, scheduling of time, arrangement of space, classification of humans, social roles and relationships, child raising, laws, rules, and more [49]. There are various definitions of "culture". Generally, culture can be understood as "that complex whole which includes knowledge, beliefs, arts, morals, laws, custom, and any other capabilities and habits acquired by a (human) as a member of society".

In 1982, social anthropologists Douglas and Wildavsky proposed "culture theory" that assumes "risk" is created within a culture [50]. Cultural theory is an approach of interpreting how and why individuals form judgements about danger and threat. The point of cultural theory is that such judgments are formed dependent on social context, and related to rights to know, justice for those likely to be affected by damage or loss of peace of mind, and about blame, responsibility and liability. Thus, what is considered as risk, and how serious a risk is thought to be being perceived differently among organizations or groups became the main focus of cultural approaches to risk [51].

Furthermore, cultural theory made valuable contributions to the deliberative process in risk management [52]. The normative implications of cultural theory in the context of risk assessment, particularly emphasizes "fairness" as the key to resolving conflicts between these different perspectives [53]. The fairness is related to how to treat different groups appropriately and may involve consensus building, and appropriate use of compensation or liability rules. A pure technical concept of risk that focuses on "how safe is safe enough" may never produce a satisfactory answer. The normative procedural recommendations of cultural theory are that "how fair is safe enough" is more important than "how safe is safe enough". Moreover, "trust" towards the institutions that create and regulate risk is of great interest [53]. Cultural theory promotes and supports direct incorporation of citizen's viewpoint in decision-making process.

3.7. Psychology

The interest in risk studies in psychology fields began with economic psychologists. Around the 1970s, a group of economic psychologists with a long experiment tradition rejected the rational choice theory of microeconomics [37]. Under the rational choice theory of economics, people are risk-neutral to optimize the expected cost or profit [54]. However, the experimental evidence demonstrated that people making judgments under uncertainty do not fully adhere to rational thought based on assessing the likelihood and severity of outcomes; yet people follow a way of intuitive thinking that makes decisions quickly and unwittingly [55–57]. This was published in the seminal work by Kahneman and Tversky, in which they demonstrated that "cognitive illusion/bias" stem from heuristics [55]. In the same work, they identified three main heuristic types, including representativeness, availability bias, adjustment and anchoring [37]. To integrate these behavioral observations into rational decision theory, Kahneman and Tversky developed an alternative risk choice model: prospect theory [58].

The other main branch of psychology studies on risk emphasizes social and cultural influence on misperception of risk. This branch of work streamed from studies on risk acceptability [59] and applied empirical psychometric approaches to quantifying risk [60]. The distinguishing feature of the psychometric approach is using various psychometric scaling methods to produce quantitative measures of perceived risk [61].

3.8. Sociology

Since the 1980s, the notion of risk has become extremely popular in many areas of social science, from policy and theory to international relations and political science. Its effect is particularly marked in sociology, where a distinctive "sociology of risk" has emerged [62]. In the mid-1980s, social scientist Beck proposed the famous "risk society" theory to replace "industrial society", arguing that risk is becoming increasingly central to our global society [63]. These risks are not only connected with industrial activities but also to how actors deal with what they perceive as risks. Risk management is no longer limited to certain technological risks but has become integral to all aspects of social life. Sociology highlights the importance of context (social, material, economic, cultural) for understanding social life. the social construction of misperception of risk, the social amplification of risk, and the social construction of risk [64]. Thus, risk management focuses not only on how organizations deal with the technical calculation of risks but also with the actors they perceive as possible threats and potential risks to the stability of the organization. Meanwhile, the urgent need for managing today's risk has promoted the creation of a new discipline that is risk assessment [60].

In 1998, Kasperson et al. proposed the Social Amplification of Risk Theory (SART) [65]. The SART framework explains how the information process, institutional structures, socialgroup behavior, and individual responses determine the consequences and how the "ripple effect" of risk events introduces the temporal and geographical extension of impacts. The SART recognizes the role of engineering and science understanding of risk, however, it argues the deficiencies of technical risk analysis cannot provide information about social indirect impacts.

3.9. Philosophy

Philosophy is distinguished by the fact that it has linkages with every other academic discipline. In the 1990s, philosophers increasingly identified a wealth of risk-related issues in need of philosophical elucidation [66]. Philosophers have illuminated new insights into the terminological clarification of risk, the ontological base and epistemic status of risk, risk and moral principles, its decision–theoretical aspects, the implications of risk in political philosophy, and various other topics [67–71]. Furthermore, the philosophers' argumentation analysis has also led risk experts in other fields to start examining the "value-laden" aspects of risk. For example, when it is ethically acceptable to expose another person to risk [68,71]? How can the definition of risk be extended to consider moral principles [72]? How can risk analysis account for the possibility that risk analysis itself is "wrong" [73]?

4. Definitions and Interpretations of Risk

After understanding the historical progression, the next step is to identify what risk means in different contexts. There is a wealth of literature discussing risk concepts. It focuses on analyzing the linkage between the definition and the disciplinary background to uncover the reasons why risk is approached differently across disciplines. To have a more coherent understanding of these definitions, four overarching categories are compared: economics and decision science, engineering and science, social science, and philosophy. We subjectively compiled a list of definitions or interpretations of risk from diverse disciplinary literature, summarized in Table 1.

Reference	Year	Descriptions, Definitions, and Understanding of Risk	Source		
[74]	1895	If there is any uncertainty as to whether or not the performance of a given act will produce a harmful result, the performance of that act is the assumption of a risk.	Social Science & Medicine		
[27]	1921	Risk is objective uncertainty.	"Risk, uncertainty and profit"		
[75]	1944	Risk is expected loss.	The Quarterly Journal of Economics		
[76]	1953	Risk is the variance of the probability distribution over the utilities of all possible consequences.	Econometrica		
[77]	1975	Risk is expected value, encompassing both the outcomes of a decision and some representation of the probability of the outcomes.	"Decision Making and Training"		
[78]	1977	Risk is the variance or dispersion of outcomes.	Journal of Accounting Research		
[79]	1981	A function of the probability of loss and the distribution of losses.	Theory and Decision		
[80]	1981	Risk is a set of triplets.	Risk Analysis		
[50]	1982	Risks are the dangers that societies define as troublesome.	Risk Analysis		
[81]	1992	A systematic way of dealing with hazards and insecurities induced and introduced by modernization itself.	Theory, Culture & Society		
[82]	1998	Risk is the relationship of responsibility for harms for which the action of some person was a condition.	RISK: Health, Safety & Environment		
[83]	1998	Risk is a situation or event where something of human value is at stake and where the outcome is uncertain.	Journal of Risk Research		
[84]	2001	The statistical probability that an undesirable event—disease or death—may strike.	Scandinavian Journal of Primary Health Care		
[85]	2001	Risk as feelings.	Psychological Bulletin		
[86]	2002	RISK = Replaces Incomplete Scientific Knowledge.	Toxicology		
[87]	2002	Risk has two components: the extent of the damage and the probability or likelihood that such damage will occur.	Safety Science		
[88]	2002	Risk is seen as a concept that human beings have invented to help them understand and cope with the dangers and uncertainties of life.	Regulation of Toxic Substances and Hazardous Waste		
[89]	2003	What is to be considered a "risk" depends entirely on cultural settings and assumptions; risks are culturally defined and selected.	Ethnos		
[90]	2005	Risk is a measure of the probability of adverse effects on health, property, and society, resulting from the exposure to a hazard of a given type and magnitude, within a certain time and area.	Natural Hazards and Earth System Sciences		
[73]	2009	Risk refers to uncertainty about and severity of the events and consequences (or outcomes) of an activity with respect to something that human value.	Journal of Risk Research		
[64]	2013	Risks are always situated in a social context and are necessarily connected to actors' activities.	"Essentials of Risk Theory"		
[91]	2015	The selection of risks reflects social and political relations within groups.	Health, Risk & Society		
[92]	2018	Risk is a state of uncertainty, s.y. there is a possibility that it involves loss or other undesirable outcome for an exposed actor.	Proceedings of the Central European Cybersecurity Conference		
[93]	2019	Risk = Hazard \times Exposure \times Susceptibility	Natural Hazards		

 Table 1. Compilation of risk definitions and interpretations from the literature.

Reference	Year	Descriptions, Definitions, and Understanding of Risk	Source		
[94]	2019	Risk = reality + objectives	Sustainability		
[95]	2019	Risk + information entropy = k	Philosophy of Science		
[96]	2020	Risk can be defined as the range of all possible, but uncertain, outcomes from positive (most optimal) to negative (most suboptimal) in terms of both reward magnitude and probability.	Current Opinion in Behavioral Sciences		

Table 1. Cont.

4.1. Economics and Decision Sciences

There are three main categories of definitions of risk in economic and decision sciences: risk = objective uncertainty, risk = variance, and risk = expected utility.

The definition of risk has evolved over time. By 1920, the dispersion of a relative frequency distribution had been clearly identified as risk [26]. However, this definition faced scrutiny since the late 19th century, particularly among authors in the insurance field [74]. They observed that not all risks have reliable historical relative frequencies or statistics. Thus, risk is still present even if quantifiable statistics are not. If risk is measured by relative frequency, then some risks are unquantifiable, which creates uncertainty. Although scholars recognized the distinction between probability and risk, few scholars are able to specify the nature of the term by definition. Willett [97] was the first scholar to separate and distinguish the various aspects of the problem of explaining the nature of risk. He thinks it is necessary to define risk with reference to the degree of uncertainty about the occurrence of a loss, not with reference to the degree of probability that it will occur. In his definition, risk is the objectified uncertainty as to the occurrence of an undesired event. However, he did not make an explanation about the term "uncertainty". Knight illuminate the ambiguities concealed in the notion of "risk" and "uncertainties" by employing the terms "objective probability" and "subjective probability" to designate the risk and uncertainty, respectively. Risk is defined as "objective uncertainty" to make a distinction with "subjective uncertainty" (uncertainty). In Knight's definition, risk could be calculated precisely with certainty (the relative frequency) [27].

Financial economics is formed on the foundation of modern portfolio theory (MPT). MPT is an investment model that describes how investors may balance risk and return in constructing investment portfolios [98]. The basic principle of risk and return trade-offs is that the riskier the investment, the greater the required potential return. In Portfolio theory, risk is synonymous with volatility, representing the chance that the actual return on an investment will be different than expected [30]. The two most common ways of measuring volatility or risk are the calculation of the variance of an expected return and the standard deviation from an expected return. It should be noted that in investment context, a decision under risk must be understood as a joint function of the expected return and variance. Sometimes, investors will choose to bear greater risks in order to achieve higher returns.

Economics and decision science risk studies are closely related to the question of "how people should make decisions". Rational choice theory is essential in modern economics. It defines that a rational individual would seek utility maximization when faced with various alternatives [57,99,100]. Normally, the premise underlying pure rational choice theory under risk is that decision-makers do not know the precise outcome but have perfect knowledge of the likelihoods of various outcomes [101]. Additionally, the EU theory acknowledges that real life scenarios are a mixed risk and uncertainty. If individuals are risk-neutral, expected values can replace the definition of probability and utility [102]. Compared with other economics definitions, it has the merit of encompassing people's subjective preference and attitude towards risk in the risk concept [103].

4.2. Engineering and Science

The quantitative definition developed by Kaplan and Garrick profoundly influences how risk is conceptualized, especially in engineering settings [80]. Engineers and scientists

prefer quantitative approaches that focus on "operationalization" and "measurement". From the engineering and science point of view, risk can be interpreted from a systemsbased perspective as a function of system states [104]. They believe that risk could be defined and measured mathematically and free from subject bias. Different from the economics system, engineering risk assessment is often associated with systems where data is insufficient or absent. Actuarial approaches are inappropriate for engineers. Risk calculations in engineering have to handle various uncertainties. Hence, experts have to judge the probabilities based on their experience and knowledge. The illustration of system-based definitions is shown in Figure 3.



Figure 3. The illustration of quantitative "set of triplets" definition of risk: $R = \{\langle s_i, l_i, x_i \rangle\}$; s_i denotes scenario *i*th risk scenario, l_i is the likelihood of the scenario, x_i denotes the potential consequences. Susceptibility/vulnerability represents the inherent states of a system to adapt and respond to hazards; threats could be seen as specific interruptions outside the system; exposure can be defined as assets and values exposed to certain threats.

There are three main discourses in defining risk in engineering and science literature: probability-based, uncertainty-based, and moral-based. This probability-based definition still plays an important role in today's engineering risk management.

The probability-based definition has two branches. One is from safety and reliability engineering, in which risk is defined as probability and consequence. In the other, risk is defined as a combination of "exposure", "susceptibility/vulnerability", and "consequence". This is because safety and reliability engineering are primarily concerned with probability of failures in the operating system itself. The second type of definition is commonly seen in environmental and security science fields. In contrast, in the field of environmental and security science, risks are caused by factors external (natural disasters, terrorist attack) to the reference system. This is what leads to the subtle difference in definition. But in "exposure", "susceptibility/vulnerability" are all calculated in the form of probability. Regardless of what variables are used to define risk, probability and the severity of consequences are two indispensable elements.

In the past decade, scientists with engineering backgrounds have paid attention to debates and discussions on the concept of risk. Scholars have focused on the distinction between risk concept and risk characterization and how to represent uncertainty accurately. They pointed out "probability" is just a way of characterizing "uncertainty", however, not the only way. Particularly, fruitful results have been yielded by introducing the "knowledge" dimension in characterizing the "uncertainty" component of risk [105,106]. The concept of risk is shift from probability-based to uncertainty-based. Furthermore, the acknowledgment of the "time" scale of risk concept provides support for resilience works [107]. According to the uncertainty-based definition, risk as a concept exists objectively and when risk is assessed it is dependent on the assessor—it becomes subjective.

Influenced by economic, science and engineering, psychometric, and social risk studies, Gardoni and Murphy proposed a synthesized risk ranking framework and applied it to dam risk assessment. The framework expands the two-dimensional risk definition to three-dimension with the addition of "source of risk" that emphasizes the moral differences between individuals. Specifically, voluntariness, distribution of consequences, and causation are incorporated to represent equity and intentionality [72].

4.3. Social Sciences

Risk does not have a precise definition in social sciences, unlike fields such as economics, decision science, engineering, and science. In social science literature, risk is a disputed concept that has comprehensive meanings. Social scientists argued that all human activity, including the physical sciences, relies on social, cultural, and political construction [108]. Based on how "risk" is constructed, different disciplines can generate different interpretations in the social sciences field.

According to anthropologists, statistical probability is a mathematical concept that quantifies the chances of an event or occurrence, and it has limited relevance for explaining how people think and act in situations where there is an element of uncertainty. For example, the cultural theory assumes that risk is a cultural phenomenon, and what constitutes "risk" is entirely dependent on cultural settings and propositions [89]. Risk is decided by a collective belief system. Thus, it varies with culture and social context.

From a psychologist's view, especially social psychology, risk could be defined by any individuals who may be influenced (e.g., experts employ risk assessment to evaluate hazards, and laypeople rely on intuitive judgments) [88,109]. However, cultural theory does not take into account individual differences in perceptions of nature. What is essential for anthropologists is to understand how groups select and frame risks. For example, some social misperception of risk studies concludes subjective attributes that influence individuals' view of acceptability of risks, voluntariness, benefits [59], dread, and familiarity [110]. The subsequent studies of risk-seeking behavior studies in economic and psychology literature also explain that risk can have a desired outcome. Depending on the value actors attributed to the manifold issues at stake, an outcome can be positive and/or negative [111].

The sociology of risk evolved from psychological and anthropological perspectives. Works on the sociology of risk generally focus how the social and cultural factors that influence the selection of "risk". In that sense, risk is also dynamic, continually modified by the societal environment. There are two major viewpoints of risk in sociology. The social construction of risk theory denies the "objective" elements of risk and regards risk as purely being shaped by social environment. Nonetheless, the SARF emphasize public perceptions can amplify the real dangers and leading to risk perception deviated from risk assessment.

In the context of law, risk is used to conceptualize and address social problems, and as an incentive to reduce individual claims on collective resources [112]. While the law demands positive statements by scientists that a risk does exist or does not exist [113], the results of risk research are often ambiguous. This makes translating scientific data into something that can be used to make decisions in a legal context difficult [114]. As such, studies in law do not emphasize probability as much as the field of engineering. Instead, the concentration of the lawyer is the causation and liability of risks [82,115].

4.4. Philosophy

Philosophy provides a new perspective for risk definition research from the perspective of epistemology and ontology. Ontology denotes the state of reality, while epistemology represents the creation of knowledge and understanding. Ontology focuses on whether risk exists objectively independent of risk observers. There are many discussions on the ontology of the concept of risk. For instance, Shrader-Frechette proposed a scientific proceduralist approach that assumes risk is neither a social construct nor purely objective [69]. Solberg and Njå maintained that all risk claims are subjective. risk as a possible state of affairs is a confusion between ontology and epistemology. They believe that when people think of risk, they always link it with specific activities or hazards. Hence, risk is a concept that we used to cope with the logically possible future states of affairs and risk does not exist

objectively [109]. Hansson [116], Blokland, and Reniers reasoned that when no value can be attributed to reality, there is nothing at risk and thus that the concept of risk is not valid [94]. Peschard et al. argued that risk is based on objective facts, but risk identification and risk estimates are value-dependent processes [117]. The epistemology of risk refers to the study of how we know and understand risk. From the epistemological aspect, risk is a state that something undesirable may happen, but we do not know whether or not it will. Risk is regard as a state of lack of scientific knowledge or information. It therefore has a strong component of "uncertainty" [118]. If uncertainty is not deemed as an element of risk, then it becomes fatalism.

5. Risk Assessment Methods

After reviewing the diverse understandings and definitions of risk, this section focuses on the different methods to measure and assess risks. We subjectively selected three standard risk assessment methods (quantitative risk assessment, multicriteria decisionmaking, and psychometric risk paradigms) and some ad hoc approaches that are tailored to specific cases or situations. Their scope of use, purpose, main methods, and advantages and some controversial points are discussed in the following context. An overview of main risk assessment methods, their advantages, limitations, and their primary fields of application is summarized in Table 2.

Table 2. Overview of main risk assessment methods, their advantages, limitations, and main application field.

Risk Assessment Method	Advantages	Limitations	Main Application Field
QRA	Help understand real damage and probabilities; Free from subject bias.	Neglect contextual issues; Data scarcity.	Engineering, environmental science, chemical engineering, energy, earth and planet science.
MCDA	Enrichment of risk assessment process; Simplicity; Transparent.	Results can be manipulated; Not suitable for time-sensitive situations; The simple algorithm may result in misleading results.	Environmental science, public policy, urban planning, energy management.
Psychometric risk paradigm	Understand public's concerns.	Public attitudes are prejudiced.	Social sciences, medicine, psychology.
Ad-hoc approach: Risk assessment framework (RAF) Person-Centered Risk Assessment Framework (PCRSF)	Comprehensiveness; Empowerment; Enhance risk awareness; Transparency.	Evaluation of these approaches are limited; Mixed information is not integrated well.	Medicine, nursing.

5.1. Quantitative Risk Assessment (QRA)

Quantitative risk assessment (QRA) is a widely used technique for identifying and estimating hazards, especially in engineering and technical fields. It relies on engineering and scientific knowledge to aid "rational" decisions when the consequences of alternatives are uncertain. It aims to provide detailed quantitative information on consequences of specific hazards under uncertainty to aid decision-making regarding risks [119]. By thoroughly examining how factors can interact, what scenarios can result, and what damage the scenarios can cause, QRA helps technicians apply quantitative approaches to handle risk aiming to provide actual values of probabilities and consequences. Through this process, QRA effectively answers three questions: "What can go wrong (scenarios)?", "How likely is it (frequency)?", and "What are the impacts?" (the process of QRA is shown in Figure 4).



Figure 4. The basic process of quantitative risk assessment.

It should be noted that QRA was first developed in the nuclear engineering context. However, according to specific characteristics in other fields of engineering and science, such as chemical risk assessment and flood risk assessment, it may have different connotations and names. For instance, QRA in chemical risk assessment involves dose–response assessment, exposure assessment, and risk characterization [120]. QRA in flood risk assessment involves hazard identification, vulnerability assessment, and exposure assessment [121]. However, the similarities of these methods are that they all utilize quantitative approaches to estimate the level of risks.

Normally, the first step of QRA involves defining the scope and context of the analyzed system. Next, the possible hazards and potential linkage between causes and consequences are identified. After that, the frequency of the consequence is analyzed. The tools used for quantitative frequency analysis include historical data, fault tree analysis (FTA), probabilistic modeling, subjective judgment, and more. The fourth step involves the analysis of the outcome of the consequence. Tools for quantitative consequence analysis include subjective judgment, event tree approach, empirical physical model, experiments, and computational simulation. Next, the estimated risk is evaluated by probability-consequence diagram, i.e., F-N curves, risk matrix [122], etc. Finally, political, procedural, and technical control and mitigation measures are identified and applied to treat risks.

There are many disputes on utilizing QRA for risk assessment. On one hand, QRA has faced many criticisms on its proficiency of science. For example, QRA is questioned not deal adequately with data-scarcity scenarios with "rare" events, where there are substantial unknown elements in quantitative risk assessment [123]. Only caring about bottom-line

numbers is not useful when they are highly uncertain, limited by risk analysts' knowledge [124], and include human error in risk estimation [125].

On the other hand, quantifications tend to reduce the complex nature of risk and cannot reflect key social, cultural, and political realities. For instance, this also leads to another criticism, as it concentrates on the direct impacts of consequences, such as physical harm to objectives, while ignoring the indirect social impacts [65]. Additionally, QRA may neglect emotional responses that importantly affect perceptions, judgments, and behaviors in response to real or perceived risks [126]. Some maintain that simpler and more popular techniques such as the precautionary principle should be used instead of or in addition to QRA [127].

Despite these, QRA does play a role in many real-world cases to help achieve preferred consequences. QRA is based on a rational idea of risk. Some studies have tried to address the limitations of risk assessment by introducing public values.

For example, Bohnenblust and Slovic proposed a framework to incorporate subjective factors such as risk aversion, willingness to pay, and voluntariness into a risk-based decision-making process [3].

$$R_m = \sum_{i=1}^n p_i C_i \varphi_i(C_i) \omega_i \tag{1}$$

where R_m is the monetary collective perceived risk, n is the number of accident scenarios i, p_i is the probability of occurrence (per year) of scenario i, C_i are the consequences (fatalities) of scenario i, $\varphi_i(C_i)$ is the risk aversion factor as a function of C_i , and ω_i is the marginal cost or willingness to pay (million SFr/fatality).

In addition, some scholars suggested including societal values and ethical principles in risk evaluation tools [128,129]. Reniers and Van Erp [130] proposed an integrated calculation formula for determining risk and obtaining risk prioritization by considering technical and non-technical factors.

$$R_i = \frac{p_i C_i^a}{\beta_i (E_i \Gamma_i^b)} \tag{2}$$

where β_i is policy factor that varies with the degree of participation in the risk due to event/scenario *i* is voluntary, E_i is a parameter representing equity principle, and F_i is a parameter representing fairness principle, *a* is aversion factor towards consequences, and *b* is factor representing the availability of combination. It contains several implicit assumptions.

5.2. Multicriteria Decision Analysis (MCDA)

Multicriteria decision analysis/making (MCDA/M) is a method that uses systematic analysis to evaluate and select alternatives based on diverse criteria to handle the uncertainty of diverse values and interests of individuals or decision-makers [131]. It is a sound analytical method for risk management where multiple and often conflicting criteria are considered. It focuses on addressing multi-faceted decision-making problems in complex scenarios, such as political or legal issues, social issues, project planning, etc.

The risk decision problem in MCDA is usually decomposed into a hierarchy of subcriteria/indicators, and each criterion/indicator is called a "risk factor" which can be analyzed independently. It focuses on comparing the "risk factor" and calculate their rankings to identify risk management priorities. Generally, a MCDA analytical framework follows the serial process, as shown in Figure 5.

It starts with identification of the risk problem. The second step aims to identify risk determinants and potential choices. In some adapted MCDA approach for risk assessment, there is an additional step, "identification of stakeholders", between Step 1 and Step 2. Stakeholders are included to gather views from them in identifying criteria and elicit priorities of the criteria. After alternatives and criteria are determined, a synthesis of knowledge and approaches, such as survey, focus group, and consultation with experts, can be used to analyze the criteria. Then, the relative importance of criteria can be obtained through

15 of 25

common approaches such as the analytical hierarchy process (AHP), outranking, etc. [132]. Finally, applying the MCDA decision rule as a tool to aid risk assessment has many advantages. First, an MCDA approach can take account into multiple dimensions of risk as well as various values of stakeholders simultaneously. Moreover, an MCDA approach can integrate various types of data. Additionally, in MCDA, all risk factors are represented using metrics, making the assessment process more transparent and accountable [133].



Figure 5. General steps of MCDA methods for risk assessment.

However, there are also a number of deficiencies of MCDA for risk assessment. One of the most crucial aspects is that the result of MCDA can be manipulated. Additionally, stakeholders' MCDA is only suitable for decision-making processes that have sufficient time. Furthermore, MCDA usually based on simple additive and multiplicative logarithms regarding criteria scores and weights. However, the proxy-based models might produce misleading results [134]. For example, in QRA, risk = probability × impact. The risk factors "probability" and "impact" are multiplicative. Then, an equivalent formula as in MCDA should be log(risk) = log(probability) + log(potential impact).

5.3. Psychometric Risk Paradigm

The "psychometric paradigm" is a method developed to understanding and quantifying laypeople's opinions and attitudes toward risks [60]. In addition to the pure aggregation of estimated likelihoods and consequences, risk assessments in psychology studies encompass more dimensions [126]. Psychologists are interested in intuitive risk judgments (perceived risk), for example, risk aversion to certain hazards, the discrepancies between experts and laypeople towards risks, and the factors behind the perceptions. Psychometric risk perception research is mainly focused on how laypersons' "perceived" risk is departure form "actual" risk. Through psychophysical scaling and multivariate analysis techniques, risk attitudes and perceptions are expressed by "cognitive maps" or quantified representations by consulting participants to rate the riskiness of hazards [60].

Fischhoff et al. hypothesized nine characteristics that influence judgments of risk perception. It includes knowledge about risk (To what extent are the risks known precisely by the persons exposed to those risks and to what extent are the risks known to science), control over risk, and the novelty of risks, among other factors. They determined that "dread" and "novelty" are the two main factors influencing people's perceptions of risk based on an examination of survey data [110].

The psychometric approach has been continuously expanded to include more perception factors. Bassarak et al. proposed that disputed risk and morality should be captured as additional dimensions of risk judgments in addition to dread and novelty [135]. Psychometric risk paradigm is widely used in various fields to study different group's perceptions of specific risks [136]. Psychometric risk studies usually use questionnaire surveys to rate possible hazards on various scales, e.g., voluntariness of risk (1 = voluntary, 5 = involuntary). Figure 6 shows an example of a psychometric risk questionnaire.

Qualitative dimensions	Description	Score (1-5)
Voluntariness	The extent to which exposure to the hazard is voluntary	
Immediacy	The extent to which the consequences are noticed immediately	
Knowledge of exposure	The extent to which a person knows if he has been exposed	
Expert knowledge	The extent to which experts know about the hazard.	
Controllability	The extent to which a victim can control the severity of	
Controllability	consequences due to exposure	
Novelty	The extent to which the hazard is new to society.	
Catastrophic potential	How many fatalities occur at once	
Dread	The extent to which the effects of exposure are dreaded	
Severity	The extent to which the consequences of exposure are severe	

Figure 6. An example of psychometric risk paradigm questionnaire.

The qualitative dimensions in the first column on the left side represent attributes or elements on how individuals perceive different risks. Those elements may vary according to the specific contexts of various hazards investigated. Each dimension is presented in a short description, providing context for respondents. Respondents are asked to rate their perception of risk on a numerical scale (in this example, from 1–5). The combination of qualitative dimensions, descriptions, and numerical scoring provides a structured way to approach individuals' assessments of different risks. Psychometric approach is effective in identifying differences and similarities among groups related to their risk attitudes and risk perceptions of specific risks. However, the results of risk perception and the approach itself are seldom directly applied in the domains of risk management.

There are many criticisms on using public risk perceptions to influence policy making. The arguments include such as public perceptions are noise/bias, public attitudes are prejudiced, managing risk merely entails managing public perceptions. The public is not homogeneous in its risk perceptions [137]; some may worry that anything can become a determiner of risk perception [138].

By contrast, the advocates for risk perception research participatory decision process. Although lay people may have little knowledge about hazards, however, their basic conceptualization of risk can be far more extensive. These perceptions reflect the real concerns of people which technological analyses often miss [139].

For example, the group of Carnegie Mellon University proposed a deliberative risk ranking method. The method advocates gathering public input to help set risk management priorities. First, a set of risk attributes are identified by experts referring to risk perception literature. Then, risk is analyzed in terms of the attributes identified. After that, the selected participants are required to rank the attributes. Finally, risk ranking is obtained, applying the multi-attribute ranking procedure [140,141].

Plattner et al. [142] built an integrated model for natural hazard risk evaluation, including scientific risk analysis, economic perspective, social environment, values, and psychological characteristics. Plattner stressed the importance of risk perception, specifically perception-affecting factor (PAF) and objective risk e_{eff} (probability times consequences), in conducting risk assessment and management. Furthermore, applying "prospect theory" in dealing with irrational thinking regarding risks is suggested. Based on expressed preference theory in psychometric studies, Plattner et al. developed a perceived risk r_{perc} formula with the integration of subjective deviation of risk perception and effective risk function e_{eff} :

$$r_{perc} = R_{eff} \frac{\sum_{i=1}^{n} (paf_i a_i)}{\sum_{i=1}^{n} a_i}$$
(3)

where paf_i [0.5, 2] is the value of the PAF, a_i [0, 1] is the weight of each paf_i , and n is the total quantity of relevant PAF. PAF includes voluntariness of risk-taking, individual reducibility of risk, knowledge and experience, endangerment, subjective damage rating, and subjective recurrence frequency perception.

5.4. Others

In addition to the three main methods mentioned above, there are other ad hoc methods that are tailored to specific situations but don't fit into conventional categories or lack formal recognition.

Risk assessment framework (RAF) is a method applied in hospitals and healthcare fields to deal with wrong medication, delayed discharge, patient claims, failure to comply with requirements, and other adverse events [143]. There are three main steps of RAF: risk identification, risk analysis, and risk treatment. The risk assessment result is summarized and documented in a risk assessment form (Figure 7).

Risk Assessment Form									
Assessor:			Ward/department		Date assessed:		Assessment no:		
1. descibe system to be assessed(aim, elements, interactions, boundary, context)									
2. Define undesired events	3.Determine contributory factors	4. Describe potential consequen ces	5. Examine current controls	6. Estimate severity	7. Estimate likelihood	8. Estimate risk level	9. Estimate risk tolerability	10. List required controls	11. Define required actions

Figure 7. Risk assessment form, adapted from [143].

Different from the other risk assessment methods we mentioned before, RAF does not convert all the aspects of risk as a common unit. Instead, it keeps the result of risk assessment in their original form. This preservation of original units enriches the understanding of risk and enhances transparency.

Furthermore, recognizing the importance of potential emotional, psychological and spiritual harms of taking away meaningful activities that contribute to quality of life, a person-centered risk assessment framework (PCRSF) is developed. This framework addresses various risks, including falls, financial abuse, physical abuse, alcohol excess, unsafe driving and medication mismanagement or inability to manage selfcare, and more situations [144] (as shown in Figure 8).

Step 3 presents a heuristic representation of indicators of psychosocial need.

It has the merit of comprehensiveness as it not only considers physical safety risk, but also take account of psychosocial factors. Furthermore, the PCRAF is designed to empower patients, allowing them to actively participate in care planning by expressing their subjective perceptions and feelings. Moreover, PCRSF can act as a communication tool to increase awareness of existing risks among clinic team members. However, the PCRAF is not widely used due to the limited capability for patient and/or caregiver to recognize risks. Some participants have expressed that completing numerous forms can be challenging and suggest that the PCRAF needs better integration into the overall assessment process.



Figure 8. Person-centered risk assessment framework (PCRSF), adapted from [144].

6. Discussion

6.1. Current Research Status

The concept of "risk" was first used in gambling and has now become part of our society. Its definition and connotation are constantly enriched with the development of society and science. There are many contradictory categories in interpretation and usage of risk, regarding objective or subjective, rational or irrational, and more. However, after reviewing the literature from various disciplines, the emergence of these controversial points is mainly due to the different starting points for looking at risks and the different context settings they made.

Engineers primarily focus on the adverse consequences of malfunctions of the technical system, often neglecting human activities within the system. In contrast, social scientists emphasize risk management, risk perception and risk communication, social acceptance of risk. Despite the presence of these contradictory elements, many socio-cultural approaches recognize the importance of techno-scientific approach to risk. The emphasis in social sciences is to broaden the scope of technical risk assessment by incorporating more societal factors, making it more meaningful to view those socio-cultural approaches to risk as an extension of risk.

In recent decades, the growing body of risk research in philosophy has introduced ethical and moral challenges to the study of technology risks, which often overlook societal dimensions [145,146]. Our rapidly changing society is rife with disputed risk problems that necessitate the inclusion of economic, moral, and ethical aspects in risk assessments [147]. Effective risk management should not only aim to reduce the frequency and negative impacts of accidents but also promote justice and democracy [148,149]. The COVID-19 pandemic has highlighted the tension between risk decision-making and public acceptance, underscoring the need to consider not just the consequences and likelihood of risks but also their distribution and the importance of building trust among the public, institutions, companies, and governments [150]. Responsibility should be prioritized as a fundamental aspect of risk management rather than a secondary concern. This calls for the development of a novel integrated risk management paradigm that takes social, ethical, political, and economic aspects into account.

Over the past three decades, numerous efforts have been made to achieve this objective. Engineers and natural scientists continue to absorb social scientists' perspectives and improve current risk assessment approaches by encompassing public values, moral principles, and utilizing deliberative process. However, these approaches have yet to be fully integrated into risk management frameworks with established procedural steps.

Recently proposed integrated risk management frameworks, such as the one developed by the International Risk Governance Council (IRGC) [151] and the SAFE FOODS framework for the governance of food safety [152], emphasize transparency, inclusive communication, stakeholder involvement, and the consideration of broader economic, social, and political contexts. These frameworks represent significant steps toward inclusive risk governance (as illustrated in Figure 9).



Figure 9. Integrated risk management framework. (a) IRGC risk governance framework, figure adapted from [151]; (b) The SAFE FOODS risk governance framework, referred to in Ref. [152].

Both risk management frameworks show an irritative decision-making process with four main stages: framing (pre-risk assessment), risk appraisal (risk-benefit assessment), risk evaluation, and risk management. They both start with pre-risk assessment/framing that focuses on framing risk and identification of involved stakeholder groups. Then, risk appraisal/risk-benefit assessment is conducted to characterize risk. However, these frameworks diverge in their focus during this stage. The IRGC risk governance framework not only includes technical risk analysis but also gathers stakeholders' concerns and opinions. In contrast, the SAFE FOODS risk governance framework emphasizes the importance of societal impacts within the risk assessment process. Following the risk appraisal, the outcomes of risk assessment are compared to specific criteria to determine the significance and acceptability of the risk. Based on the results of risk evaluation, tolerable risks are then subjected to treatment. Despite these structured processes, several unresolved issues persist within both frameworks.

- How can public concerns be effectively measured or assessed, and in what ways do
 these concerns influence policymaking?
- What methods can be employed to rank risks while taking into account non-rational factors?
- How should decisions be made when conflicts arise between technical risk analysis assessments of public concerns?

6.2. Recommendations for Future Works

Based on the literature review results and current research status, some recommendations for future research are given.

6.2.1. Expanding Risk Dimensions for More Informed Decision Making

Social sciences and philosophy studies help to identify and explain public concern, economic, and psychological factors. A number of non-rational elements have been identified. For example, emotions such as voluntariness, distribution of risks and benefits, available alternatives [153], fear, happiness, anger, and surprise are positively correlated with risk aversion [154]. However, the results of these studies have rarely been applied in the formal risk management model. Methods of selecting and combining these aspects with the current technical concepts of risk need to be explored.

In future studies, it is necessary to expand the definitions of risk instead of focusing on one or two aspects to better cope and manage risks.

6.2.2. Need Integrated Approaches to Risk Assessment

Prior studies have proposed various methods for assessing concerns and ethical issues separately as a supplement to the main risk assessment process. These qualitative assessments based on expert judgment are also called concern-driven risk management [127]. For example, Hansson [155] introduced ethical risk analysis (ERA) to assess ethical related decisions on risk. The method covers ethical issues such as agency, interpersonal relationships, and justice. Malsch et al. [156] developed an innovative safety decision support system with the integration of ethical impact assessment into risk governance framework, in which social (e.g., equity, gender), economic, and environment-related values are included. Additionally, the UK Treasury method proposed additional scale-based assessment methods to take public concerns, such as familiarity, trust, dread, equity, and management, into account [157].

Generally, these contextual factors are assessed using non-quantitative methods. Given that quantitative approaches dominate current risk management frameworks, further investigation is needed to explore how to synthesize various types of information and data to aid decision-making.

6.2.3. Lack of Normative Procedures of Stakeholder Involvement in Risk Management

The cultural approach, psychometric approach, and moral philosophy all advocate people who may affected by risk should have input in risk management. A fair decision-making process that incorporates stakeholder participation can lead to procedural justice [158,159]. Stakeholder pressure is a significant factor influencing risk management strategies [160]. Over the past decade, there has been a growing interest in involving stakeholders in natural hazard risk management in Europe, particularly concerning flood risk management [161,162].

While there have been attempts to introduce stakeholders in the MCDA method, the approaches vary significantly from study to study. Despite many risk-related standards stressing the importance of stakeholders, systematic procedures for integrating stakeholders in risk management remain lacking [163]. A normative model that can systematically incorporate stakeholder involvement is necessary. Furthermore, scholars should explore ways to better manage diverse stakeholder opinions regarding risk, the interrelations between risk management and stakeholder management, and stakeholders' responsibility and roles in the risk management process.

7. Conclusions

This paper presents a comprehensive review of the historical development, theories, approaches of risk concept used in domain sciences when studying, discussing, and managing risks. The main conclusions are as follows.

 There are wide variations and even competencies between and within disciplines regarding the notion of risk. However, many of the contradictive perspectives between the concept of risk are due to different standpoints. From a holistic point of view, the social science approach extends the concept of risk. The concept of risk in modern risk science has a deep technical background that emphasizes the scientific nature of risk. However, risk resulting from a technical system is not isolated and may bring indirect impacts to the broader social system. Risk can not only be identified through scientific methods but also constructed from social activities.

2. The concept of risk has never evolved in isolation; its development has been shaped by the integration of theories from various disciplines. For instance, early probability theory in mathematics has long served as the foundation for quantitative risk assessment. In recent years, more and more risk management studies tried to integrate "hard" and "soft" approaches and provide meaningful results. However, a normative procedural integrated risk management framework still needs to be explored.

To conclude, a comprehensive risk perspective with the integration of diverse values, including technical aspects, social and cultural response, individual cognitive processes, etc., may have the potential to deal with risks in a sustainable and responsible way.

Author Contributions: Conceptualization, Y.X. and G.R.; methodology, Y.X., GR. and M.Y.; writing—original draft preparation, Y.X.; writing—review and editing, G.R. and M.Y.; visualization, Y.X.; supervision, G.R. and M.Y. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Acknowledgments: This work is supported by the China Scholarship Council (Grant No. 202106340031).

Conflicts of Interest: The authors declare no conflicts of interest.

References

- 1. Athearn, J.L. What is risk? J. Risk Insur. 1971, 38, 639–645. [CrossRef]
- 2. Greenberg, M.; Cox, A.; Bier, V.; Lambert, J.; Lowrie, K.; North, W.; Wu, F. Risk analysis: Celebrating the accomplishments and embracing ongoing challenges. *Risk Anal.* 2020, 40, 2113–2127. [CrossRef] [PubMed]
- Bohnenblust, H.; Slovic, P. Integrating technical analysis and public values in risk-based decision making. *Reliab. Eng. Syst. Saf.* 1998, 59, 151–159. [CrossRef]
- 4. Merkelsen, H. The constitutive element of probabilistic agency in risk: A semantic analysis of risk, danger, chance, and hazard. *J. Risk Res.* **2011**, *14*, 881–897. [CrossRef]
- 5. Aven, T. The risk concept—Historical and recent development trends. Reliab. Eng. Syst. Saf. 2012, 99, 33–44. [CrossRef]
- 6. Boholm, M. How do Swedish Government agencies define risk? J. Risk Res. 2019, 22, 717–734. [CrossRef]
- 7. Covello, V.T.; Mumpower, J. Risk analysis and risk management: An historical perspective. Risk Anal. 1985, 5, 103–120. [CrossRef]
- 8. Beck, U. Living in the world risk society: A Hobhouse Memorial Public Lecture given on Wednesday 15 February 2006 at the London School of Economics. *Econ. Soc.* 2006, *35*, 329–345. [CrossRef]
- 9. Jasanoff, S. Bridging the two cultures of risk analysis. Risk Anal. 1993, 13, 123–129. [CrossRef]
- 10. Althaus, C.E. A disciplinary perspective on the epistemological status of risk. Risk Anal. Int. J. 2005, 25, 567–588. [CrossRef]
- 11. Renn, O. Concepts of risk: An interdisciplinary review part 1: Disciplinary risk concepts. *GAIA-Ecol. Perspect. Sci. Soc.* 2008, 17, 50–66. [CrossRef]
- 12. Renn, O. Concepts of Risk: A Classification; University of Stuttgart: Stuttgart, Germany, 1992.
- 13. Committee of Sponsoring Organizations. *Enterprise Risk Management Framework;* Committee of Sponsoring Organizations (COSO): New York, NY, USA, 2020.
- 14. National Institute for Public Health and the Environment. *National Risk Assessment of the Kingdom of the Netherlands* 2022; National Institute for Public Health and the Environment: Utrecht, The Netherlands, 2023.
- ISO 31000; Risk Management—Principles and Guidelines. International Organization for Standardization: Geneva, Switzerland, 2018.
 UK Cabinet Office. *Risk: Improving Government's Capability to Handle Risk and Uncertainty;* Strategy Unit Report; UK Cabinet Office:
- OK Cabillet Office. Risk. Improving Government's Capability to Hundle Risk and Officertainty, Strategy Offic Report, OK Cabillet Office. London, UK, 2002.
- 17. Ale, B.; Aven, T.; Jongejan, R. Review and discussion of basic concepts and principles in integrated risk management. In *Reliability, Risk, and Safety, Three Volume Set;* CRC Press: Boca Raton, FL, USA, 2009; pp. 455–462.
- Wohlin, C. Guidelines for snowballing in systematic literature studies and a replication in software engineering. In Proceedings of the 18th International Conference on Evaluation and Assessment in Software Engineering, London, UK, 13–14 May 2014; pp. 1–10.
- 19. Jacobs, L.A. An analysis of the concept of risk. *Cancer Nurs.* 2000, 23, 12–19. [CrossRef] [PubMed]
- 20. Dein, S. The anthropology of uncertainty: Magic, witchcraft and risk and forensic implications. *J. Forens. Anthropol.* **2016**, *1*, 107. [CrossRef]
- 21. Boholm, A. Anthropology and Risk; Routledge: Abingdon, UK, 2015.
- 22. Peterson, M. Expected utility. In Nonbayesian Decision Theory: Beliefs and Desires as Reasons for Action; Springer: Berlin/Heidelberg, Germany, 2008; pp. 109–125.

- 23. Schoemaker, P.J. The expected utility model: Its variants, purposes, evidence and limitations. J. Econ. Lit. 1982, 20, 529–563.
- 24. Dannreuther, C.; Lekhi, R. Globalization and the political economy of risk. *Rev. Int. Political Econ.* 2000, 7, 574–594. [CrossRef]
- 25. Wood, O.G. Evolution of the concept of risk. J. Risk Insur. 1964, 31, 83–91. [CrossRef]
- 26. McGoun, E.G. The history of risk "measurement". Crit. Perspect. Account. 1995, 6, 511–532. [CrossRef]
- 27. Knight Frank, H. *Risk, Uncertainty and Profit;* Liberty Fund: Carme, IN, USA, 1921.
- 28. Gollier, C. The Economics of Risk and Time; MIT Press: Cambridge, MA, USA, 2001.
- 29. Broome, J. Utility. Econ. Philos. 1991, 7, 1–12. [CrossRef]
- 30. Mangram, M.E. A simplified perspective of the Markowitz portfolio theory. Glob. J. Bus. Res. 2013, 7, 59–70.
- 31. Tucker, B.; Nelson, D.R. What does economic anthropology have to contribute to studies of risk and resilience? *Econ. Anthropol.* **2017**, *4*, 161–172. [CrossRef]
- 32. Zio, E. Reliability engineering: Old problems and new challenges. Reliab. Eng. Syst. Saf. 2009, 94, 125–141. [CrossRef]
- Zachmann, K. Risk in historical perspective: Concepts, contexts, and conjunctions. In *Risk-A Multidisciplinary Introduction*; Springer: Berlin/Heidelberg, Germany, 2014; pp. 3–35.
- 34. Renn, O. Three decades of risk research: Accomplishments and new challenges. J. Risk Res. 1998, 1, 49–71. [CrossRef]
- 35. Dionne, G. Risk management: History, definition, and critique. Risk Manag. Insur. Rev. 2013, 16, 147–166. [CrossRef]
- 36. Rasmussen, N.C. *Reactor Safety Study: An Assessment of Accident Risks in US Commercial Nuclear Power Plants;* National Technical Information Service: Alexandria, VA, USA, 1974; Volume 7.
- 37. Zinn, J.O.; Taylor-Gooby, P. Risk as an interdisciplinary research area. Risk Soc. Sci. 2006, 1, 20–53.
- 38. Nateghi, R.; Aven, T. Risk analysis in the age of big data: The promises and pitfalls. Risk Anal. 2021, 41, 1751–1758. [CrossRef]
- 39. Bradbury, J.A. The policy implications of differing concepts of risk. Sci. Technol. Hum. Values 1989, 14, 380–399. [CrossRef]
- 40. Rørtveit, G.; Strand, R. Risk, uncertainty and ignorance in medicine. Tidsskr. Den Nor. Legeforening 2001, 121, 1382–1386.
- 41. Hayes, M.V. On the epistemology of risk: Language, logic and social science. Soc. Sci. Med. 1992, 35, 401-407. [CrossRef]
- 42. Skolbekken, J.A. The risk epidemic in medical journals. Soc. Sci. Med. 1995, 40, 291–305. [CrossRef]
- 43. Alfieri, A.V. The fall of legal ethics and the rise of risk management. *Geo. LJ* 2005, 94, 1909.
- 44. Priest, G.L. The new legal structure of risk control. Daedalus 1990, 119, 207-227.
- 45. Pratt, J.; Miao, M. Risk, Populism, and Criminal Law. New Crim. Law Rev. 2019, 22, 391–433. [CrossRef]
- 46. MacAlister, D. Use of risk assessments by Canadian judges in the determination of dangerous and long-term offender status, 1997–2002. In *Law and Risk*; University of British Columbia Press: Vancouver, BC, Canada, 2005; pp. 20–50.
- 47. Faas, A.J.; Barrios, R.E. Applied anthropology of risk, hazards, and disasters. Hum. Organ. 2015, 74, 287–295. [CrossRef]
- 48. Velimirovic, H. An Anthropological View of Risk Phenomena; IIASA: Laxenburg, Austria, 1975.
- Hoffman, S.M. Culture: The crucial factor in hazard, risk, and disaster recovery: The anthropological perspective. In *Hazards*, *Risks, and Disasters in Society*; Academic Press: Cambridge, MA, USA, 2015; pp. 289–305.
- 50. Douglas, M.; Wildavsky, A. How can we know the risks we face? Why risk selection is a social process 1. *Risk Anal.* **1982**, *2*, 49–58. [CrossRef]
- Lupton, D. (Ed.) *Risk and Sociocultural Theory: New Directions and Perspectives;* Cambridge University Press: Cambridge, UK, 1999.
 Renn, O. The challenge of integrating deliberation and expertise: Participation and discourse in risk management. In *Risk Analysis*
- and Society: An Interdisciplinary Characterization of the Field; Cambridge University Press: Cambridge, UK, 2004; pp. 289–366.
- 53. Tansey, J.; O'riordan, T. Cultural theory and risk: A review. Health Risk Soc. 1999, 1, 71–90. [CrossRef]
- 54. Jammernegg, W.; Kischka, P. Risk-averse and risk-taking newsvendors: A conditional expected value approach. *Rev. Manag. Sci.* **2007**, *1*, 93–110. [CrossRef]
- 55. Tversky, A.; Kahneman, D. Judgment under Uncertainty: Heuristics and Biases: Biases in judgments reveal some heuristics of thinking under uncertainty. *Science* **1974**, *185*, 1124–1131. [CrossRef]
- 56. Smith, V.L. Rational choice: The contrast between economics and psychology. J. Political Econ. 1991, 99, 877–897. [CrossRef]
- 57. Mathis, K.; Steffen, A.D. From rational choice to behavioural economics. In *European Perspectives on Behavioural Law and Economics*; Springer: Cham, Germany, 2015; pp. 31–48.
- 58. Kahneman, D.; Tversky, A. On the interpretation of intuitive probability: A reply to Jonathan Cohen. *Cognition* **1979**, *7*, 409–411. [CrossRef]
- 59. Starr, C. Social benefit versus technological risk: What is our society willing to pay for safety. *Science* **1969**, *165*, 1232–1238. [CrossRef]
- 60. Slovic, P. Perception of risk. Science 1987, 236, 280–285. [CrossRef] [PubMed]
- 61. Slovic, P. Perception of Risk: Reflections on the Psychometric Paradigm; Praeger: Westport, CT, USA, 1992.
- 62. Reith, G. Uncertain times: The notion of 'risk' and the development of modernity. Time Soc. 2004, 13, 383–402. [CrossRef]
- 63. Beck, U. Risikogesellschaft: Auf Dem Weg in Eine Andere Moderne; Suhrkamp: Frankfurt am Main, Germany, 1986.
- 64. Lidskog, R.; Sundqvist, G. Sociology of risk. In Essentials of Risk Theory; Springer: Dordrecht, The Netherlands, 2013; pp. 75–105.
- 65. Kasperson, R.E.; Renn, O.; Slovic, P.; Brown, H.S.; Emel, J.; Goble, R.; Ratick, S. The social amplification of risk: A conceptual framework. *Risk Anal.* **1988**, *8*, 177–187. [CrossRef]
- 66. Cranor, C.F. Some moral issues in risk assessment. *Ethics* **1990**, *101*, 123–143. [CrossRef]
- 67. Hansson, S.O. A Panorama of the Philosophy of Risk; Springer: Dordrecht, The Netherlands, 2012.
- 68. Ersdal, G.; Aven, T. Risk informed decision-making and its ethical basis. Reliab. Eng. Syst. Saf. 2008, 93, 197–205. [CrossRef]

- 69. Shrader-Frechette, K. Technology and Ethics. In Technology and Values: Essential Readings; Wiley-Blackwell: Hoboken, NJ, USA, 2009.
- 70. Roeser, S.; Hillerbrand, R.; Sandin, P.; Peterson, M. (Eds.) *Handbook of Risk Theory: Epistemology, Decision Theory, Ethics, and Social Implications of Risk*; Springer: Dordrecht, The Netherlands, 2012; Volume 1.
- 71. Vanem, E. Ethics and fundamental principles of risk acceptance criteria. Saf. Sci. 2012, 50, 958–967. [CrossRef]
- 72. Gardoni, P.; Murphy, C. A scale of risk. Risk Anal. 2014, 34, 1208–1227. [CrossRef]
- 73. Aven, T.; Renn, O. On risk defined as an event where the outcome is uncertain. J. Risk Res. 2009, 12, 1–11. [CrossRef]
- 74. Haynes, J. Risk as an economic factor. Q. J. Econ. 1895, 9, 409–449. [CrossRef]
- 75. Domar, E.D.; Musgrave, R.A. Proportional income taxation and risk-taking. Q. J. Econ. 1944, 58, 388–422. [CrossRef]
- 76. Allais, M. Le Comportement de l'homme rationnel devant le risque: Critique des postulats et axiomes de l'école américaine. *Econom. XXI* **1953**, *21*, 503–546.
- 77. Nickerson, R.S.; Feehrer, C.E. Decision Making and Training: A Review of Theoretical and Empirical Studies of Decision Making and Their Implications for the Training of Decision Makers; Bolt Beranek and Newman, Inc.: Cambridge, MA, USA, 1975.
- Libby, R.; Fishburn, P.C. Behavioral models of risk taking in business decisions: A survey and evaluation. J. Account. Res. 1977, 15, 272–292. [CrossRef]
- 79. Fishburn, P.C. Subjective expected utility: A review of normative theories. Theory Decis. 1981, 13, 139–199. [CrossRef]
- 80. Kaplan, S.; Garrick, B.J. On the quantitative definition of risk. Risk Anal. 1981, 1, 11–27. [CrossRef]
- 81. Beck, U. From industrial society to the risk society: Questions of survival, social structure and ecological enlightenment. *Theory Cult. Soc.* **1992**, *9*, 97–123. [CrossRef]
- 82. Rehmann-Sutter, C. Involving others: Towards an ethical concept of risk. Risk 1998, 9, 119.
- 83. Rosa, E.A. Metatheoretical foundations for post-normal risk. J. Risk Res. 1998, 1, 15–44. [CrossRef]
- 84. Reventlow, S.; Hvas, A.C.; Tulinius, C. In really great danger? The concept of risk in general practice. *Scand. J. Prim. Health Care* **2001**, *19*, 71–75.
- 85. Loewenstein, G.F.; Weber, E.U.; Hsee, C.K.; Welch, N. Risk as feelings. Psychol. Bull. 2001, 127, 267. [CrossRef]
- 86. Malmfors, T.; Rosing, H. Introduction—Risk from a philosophy of science point of view. *Toxicology* **2002**, *181*, 109–113. [CrossRef]
- 87. Ale, B.J.M. Risk assessment practices in The Netherlands. Saf. Sci. 2002, 40, 105–126. [CrossRef]
- 88. Slovic, P.; Weber, E.U. Perception of risk posed by extreme events. In *Regulation of Toxic Substances and Hazardous Waste*, 2nd ed.; Applegate, J.S., Laitos, J.G., Gaba, J.M., Sachs, N.M., Eds.; Foundation Press: Goleta, CA, USA, 2002.
- 89. Boholm, Å. The cultural nature of risk: Can there be an anthropology of uncertainty? Ethnos 2003, 68, 159–178. [CrossRef]
- 90. Hufschmidt, G.; Crozier, M.; Glade, T. Evolution of natural risk: Research framework and perspectives. *Nat. Hazards Earth Syst. Sci.* 2005, *5*, 375–387. [CrossRef]
- 91. Alaszewski, A. Anthropology and risk: Insights into uncertainty, danger and blame from other cultures–A review essay. *Health Risk Soc.* **2015**, *17*, 205–225. [CrossRef]
- 92. Goman, M. Towards unambiguous IT risk definition. In Proceedings of the Central European Cybersecurity Conference 2018, Ljubljana, Slovenia, 15–16 November 2018; pp. 1–6.
- 93. Prabnakorn, S.; Maskey, S.; Suryadi, F.X.; de Fraiture, C. Assessment of drought hazard, exposure, vulnerability, and risk for rice cultivation in the Mun River Basin in Thailand. *Nat. Hazards* **2019**, *97*, 891–911. [CrossRef]
- 94. Blokland, P.; Reniers, G. An ontological and semantic foundation for safety and security science. *Sustainability* **2019**, *11*, 6024. [CrossRef]
- 95. Babic, B. A theory of epistemic risk. Philos. Sci. 2019, 86, 522–550. [CrossRef]
- 96. Freeland, C.M.; Knes, A.S.; Robinson, M.J.F. Translating concepts of risk and loss in rodent models of gambling and the limitations for clinical applications. *Curr. Opin. Behav. Sci.* 2020, *31*, 76–82. [CrossRef]
- 97. Willett, A.H. The Economic Theory of Risk and Insurance; Columbia University Press: New York, NY, USA, 1901.
- 98. Holton, G.A. Defining risk. Financ. Anal. J. 2004, 60, 19–25. [CrossRef]
- 99. Weirich, P. Expected utility and risk. Br. J. Philos. Sci. 1986, 37, 419-442. [CrossRef]
- 100. Sugden, R. Rational choice: A survey of contributions from economics and philosophy. Econ. J. 1991, 101, 751–785. [CrossRef]
- 101. March, J.G. Primer on Decision Making: How Decisions Happen; Simon and Schuster: Manhattan, NY, USA, 1994.
- 102. Tversky, A.; Kahneman, D. Advances in prospect theory: Cumulative representation of uncertainty. *J. Risk Uncertain.* **1992**, *5*, 297–323. [CrossRef]
- 103. Fishburn, P.C. Utility theory. Manag. Sci. 1968, 14, 335–378. [CrossRef]
- 104. Haimes, Y.Y. On the complex definition of risk: A systems-based approach. Risk Anal. Int. J. 2009, 29, 1647–1654. [CrossRef]
- 105. Aven, T.; Reniers, G. How to define and interpret a probability in a risk and safety setting. Saf. Sci. 2013, 51, 223–231. [CrossRef]
- 106. Aven, T.; Renn, O.; Rosa, E.A. On the ontological status of the concept of risk. Saf. Sci. 2011, 49, 1074–1079. [CrossRef]
- 107. Logan, T.M.; Aven, T.; Guikema, S.D.; Flage, R. Risk science offers an integrated approach to resilience. *Nat. Sustain.* 2022, *5*, 741–748. [CrossRef]
- 108. Douglas, M. The depoliticization of risk. In Culture Matters; Routledge: Abingdon, UK, 2018; pp. 121–132.
- 109. Solberg, Ø.; Njå, O. Reflections on the ontological status of risk. J. Risk Res. 2012, 15, 1201–1215. [CrossRef]
- 110. Fischhoff, B.; Slovic, P.; Lichtenstein, S.; Read, S.; Combs, B. How safe is safe enough? A psychometric study of attitudes towards technological risks and benefits. *Policy Sci.* **1978**, *9*, 127–152. [CrossRef]

- 111. Battistelli, F.; Galantino, M.G. Dangers, risks and threats: An alternative conceptualization to the catch-all concept of risk. *Curr. Sociol.* **2019**, *67*, 64–78. [CrossRef]
- 112. Baker, T.; Simon, J. (Eds.) *Embracing Risk: The Changing Culture of Insurance and Responsibility*; University of Chicago Press: Chicago, IL, USA, 2002.
- 113. Foster, K.R.; Bernstein, D.E.; Huber, P.W.; Huber, P.W. (Eds.) *Phantom Risk: Scientific Inference and the Law*; MIT Press: Cambridge, MA, USA, 1999.
- 114. Heilbrun, K. Prediction versus management models relevant to risk assessment: The importance of legal decision-making context. *Law Hum. Behav.* **1997**, *21*, 347–359. [CrossRef] [PubMed]
- 115. Valverde, M.; Levi, R.; Moore, D. Legal knowledges of risk. In *Law and Risk*; University of British Columbia Press: Vancouver, BC, Canada, 2005; pp. 86–120.
- 116. Hansson, S.O. Risk: Objective or subjective, facts or values. J. Risk Res. 2010, 13, 231–238. [CrossRef]
- 117. Peschard, I.; Benétreau-Dupin, Y.; Wessels, C. Philosophy and Science of Risk: An Introduction; Taylor & Francis: Oxford, UK, 2022.
- 118. Hansson, S.O. Uncertainties in the knowledge society. Int. Soc. Sci. J. 2022, 54, 39–46. [CrossRef]
- 119. Arendt, J.S. Using quantitative risk assessment in the chemical process industry. *Reliab. Eng. Syst. Saf.* **1990**, *29*, 29133–29149. [CrossRef]
- 120. Haas, C.N.; Rose, J.B.; Gerba, C.; Regli, S. Risk assessment of virus in drinking water. Risk Anal. 1993, 13, 545–552. [CrossRef]
- 121. Apel, H.; Thieken, A.H.; Merz, B.; Blöschl, G. Flood risk assessment and associated uncertainty. *Nat. Hazards Earth Syst. Sci.* 2004, *4*, 295–308. [CrossRef]
- 122. Ale, B.; Burnap, P.; Slater, D. On the origin of PCDS-(Probability consequence diagrams). Saf. Sci. 2015, 72, 229-239. [CrossRef]
- 123. Weinberg, A.M. Reflections on risk assessment. Risk Anal. 1981, 1, 5–7. [CrossRef]
- 124. Apostolakis, G.E. How Useful Is Quantitative Risk Assessment? Risk Anal. Int. J. 2004, 24, 515–520. [CrossRef]
- 125. Freudenburg, W.R. Perceived risk, real risk: Social science and the art of probabilistic risk assessment. *Science* **1988**, 242, 44–49. [CrossRef]
- 126. Slovic, P.E. The Perception of Risk; Earthscan Publications: Oxford, UK, 2000.
- Anthony (Tony) Cox, L., Jr. Does concern-driven risk management provide a viable alternative to QRA? *Risk Anal. Int. J.* 2007, 27, 27–43. [CrossRef] [PubMed]
- 128. Johansen, I.L.; Rausand, M. Foundations and choice of risk metrics. Saf. Sci. 2014, 62, 386–399. [CrossRef]
- 129. Strouth, A.; McDougall, S. Societal risk evaluation for landslides: Historical synthesis and proposed tools. *Landslides* **2021**, *18*, 1071–1085. [CrossRef]
- Reniers, G.L.; Van Erp, H.N. Operational Safety Economics: A Practical Approach Focused on the Chemical and Process Industries; John Wiley & Sons: Hoboken, NJ, USA, 2016.
- 131. Kiker, G.A.; Bridges, T.S.; Varghese, A.; Seager, T.P.; Linkov, I. Application of multicriteria decision analysis in environmental decision making. *Integr. Environ. Assess. Manag. Int. J.* 2005, *1*, 95–108. [CrossRef]
- 132. Saaty, T.L. The Analytic Hierarchy Process; MCGraw Hill: New York, NY, USA, 1980.
- 133. Hongoh, V.; Hoen, A.G.; Aenishaenslin, C.; Waaub, J.P.; Bélanger, D.; Michel, P.; Lyme-MCDA Consortium lymemcda@ gmail.com. Spatially explicit multi-criteria decision analysis for managing vector-borne diseases. *Int. J. Health Geogr.* **2011**, *10*, 70. [CrossRef]
- 134. Keisler, J.M.; Linkov, I. Use and Misuse of MCDA to Support Decision Making Informed by Risk. *Risk Anal.* 2021, 41, 1513–1521. [CrossRef]
- 135. Bassarak, C.; Pfister, H.R.; Böhm, G. Dispute and morality in the perception of societal risks: Extending the psychometric model. *J. Risk Res.* 2017, 20, 299–325. [CrossRef]
- 136. Jenkin, C.M. Risk perception and terrorism: Applying the psychometric paradigm. Homel. Secur. Aff. 2006, 2, 2.
- 137. Pidgeon, N. Risk assessment, risk values and the social science programme: Why we do need risk perception research. *Reliab. Eng. Syst. Saf.* **1998**, *59*, 5–15. [CrossRef]
- 138. Slovic, P.; Fischhoff, B.; Lichtenstein, S. Why study risk perception? Risk Anal. 1982, 2, 83–93. [CrossRef]
- 139. Renn, O. Concepts of risk: An interdisciplinary review–part 2: Integrative approaches. *GAIA-Ecol. Perspect. Sci. Soc.* 2008, 17, 196–204. [CrossRef]
- 140. Florig, H.K.; Morgan, M.G.; Morgan, K.M.; Jenni, K.E.; Fischhoff, B.; Fischbeck, P.S.; DeKay, M.L. A deliberative method for ranking risks (I): Overview and test bed development. *Risk Anal.* **2001**, *21*, 913–921. [CrossRef] [PubMed]
- 141. Morgan, K.M.; DeKay, M.L.; Fischbeck, P.S.; Morgan, M.G.; Fischhoff, B.; Florig, H.K. A deliberative method for ranking risks (II): Evaluation of validity and agreement among risk managers. *Risk Anal.* **2001**, *21*, 923–937. [CrossRef] [PubMed]
- 142. Plattner, T.; Plapp, T.; Hebel, B. Integrating public risk perception into formal natural hazard risk assessment. *Nat. Hazards Earth Syst. Sci.* **2006**, *6*, 471–483. [CrossRef]
- 143. Kaya, G.K.; Ward, J.R.; Clarkson, P.J. A framework to support risk assessment in hospitals. *Int. J. Qual. Health Care* 2019, 31, 393–401. [CrossRef]
- 144. Lee, L.; Hillier, L.M.; Lu, S.K.; Martin, S.D.; Pritchard, S.; Janzen, J.; Slonim, K. Person-centered risk assessment framework: Assessing and managing risk in older adults living with dementia. *Neurodegener. Dis. Manag.* **2019**, *9*, 47–57. [CrossRef]
- 145. Asveld, L.; Roeser, S. (Eds.) *The Ethics of Technological Risk*; Earthscan: London, UK, 2009.
- 146. Van de Poel, I. An ethical framework for evaluating experimental technology. Sci. Eng. Ethics 2016, 22, 667–686. [CrossRef]
- 147. Reniers, G. On the future of safety in the manufacturing industry. Procedia Manuf. 2017, 13, 1292–1296. [CrossRef]

- 148. Modarres, M. What Every Engineer Should Know About Reliability and Risk Analysis; CRC Press: Boca Raton, FL, USA, 2018.
- 149. Meyer, T.; Reniers, G. Engineering risk management. In *Engineering Risk Management*; De Gruyter: Berlin, Germany, 2022.
- 150. Ale, B.J.; Slater, D.H.; Hartford, D.N. The ethical dilemmas of risky decisions. *Risk Anal.* 2023, 43, 219–233. [CrossRef]
- 151. Islamic Revolutionary Guard Corps. Introduction to the IRGC Risk Governance Framework, Revised Version; EPFL International Risk Governance Center: Lausanne, Switzerland, 2017.
- 152. Knudsen, I.B. The SAFE FOODS framework for integrated risk analysis of food: An approach designed for science-based, transparent, open and participatory management of food safety. *Food Control* **2010**, *21*, 1653–1661. [CrossRef]
- 153. Roeser, S. The role of emotions in judging the moral acceptability of risks. Saf. Sci. 2006, 44, 689–700. [CrossRef]
- 154. Nguyen, Y.; Noussair, C.N. Risk aversion and emotions. Pac. Econ. Rev. 2014, 19, 296–312. [CrossRef]
- 155. Hansson, S.O. How to perform an ethical risk analysis (eRA). Risk Anal. 2018, 38, 1820–1829. [CrossRef] [PubMed]
- 156. Malsch, B.; Tremblay, M.S.; Cohen, J. Non-audit engagements and the creation of public value: Consequences for the public interest. *J. Bus. Ethics* **2022**, *178*, 467–479. [CrossRef]
- 157. Treasury HM, S. Managing Risks to the Public: Appraisal Guidance; HM Treasury: London, UK, 2005.
- 158. Begg, C. Power, responsibility and justice: A review of local stakeholder participation in European flood risk management. *Local Environ.* **2018**, *23*, 383–397. [CrossRef]
- 159. Renn, O. Stakeholder and public involvement in risk governance. Int. J. Disaster Risk Sci. 2015, 6, 8–20. [CrossRef]
- 160. Betts, T.K.; Wiengarten, F.; Tadisina, S.K. Exploring the impact of stakeholder pressure on environmental management strategies at the plant level: What does industry have to do with it? *J. Clean. Prod.* **2015**, *92*, 282–294. [CrossRef]
- Evers, M.; Jonoski, A.; Maksimovič, Č.; Lange, L.; Ochoa Rodriguez, S.; Teklesadik, A.; Makropoulos, C. Collaborative modelling for active involvement of stakeholders in urban flood risk management. *Nat. Hazards Earth Syst. Sci.* 2012, 12, 2821–2842. [CrossRef]
- 162. Buchecker, M.; Salvini, G.; Di Baldassarre, G.; Semenzin, E.; Maidl, E.; Marcomini, A. The role of risk perception in making flood risk management more effective. *Nat. Hazards Earth Syst. Sci.* **2013**, *13*, 3013–3030. [CrossRef]
- 163. IRGC. Involving Stakeholders in the Risk Governance Process; EPFL International Risk Governance Center: Lausanne, Switzerland, 2020.

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.